Author's response to reviews

Title: Chemical profile and in vivo hypoglycemic effects of Syzygium jambos, Costus speciosus and Tapeinochilos ananassae plant extracts used as diabetes adjuvants in Puerto Rico

Authors:

- Jannette Gavillán-Suárez (jannette.gavillan@upr.edu)
- Alexandra Aguilar-Pérez (aapoadrgs@gmail.com)
- Natalie Rivera-Ortiz (natalie_rivera_ortiz@baxter.com)
- Karla Rodriguez-Tirado (karla.rodriguez5@upr.edu)
- Wanda Figueroa-Cuilán (wmffn8@mail.missouri.edu)
- Lorelein Morales-Santiago (lorelein.morales@upr.edu)
- Gerónimo Maldonado-Martínez (geronimo.maldonado@gmail.com)
- Luis A. Cubano (lacoardgs@gmail.com)
- Michelle M. Martínez-Montemayor (mmtz92@gmail.com)

Version: 5
Date: 17 June 2015

Author's response to reviews:

May 26, 2015

Tom Rowles, Ph.D.
Senior Executive Editor
BMC Complementary and Alternative Medicine

Dear Dr. Rowles:

We deeply appreciate the questions and comments of the reviewers during the analysis of our manuscript MS 1222172774159583. We hope that our revisions and answers fulfill the reviewer’s assessment and make our manuscript suitable for publication. This work is unique and describes the validation of herbal remedies used as diabetes adjuvants by communities in the Caribbean.

Below, we have included the reviewer’s comments followed by our point-by-point response to their concerns and a description of the text changed in the manuscript to make the content more descriptive and accurate.

Title: Chemical profile and in vivo hypoglycemic effects of Syzygium jambos, Costus speciosus and Tapeinochilos ananassae plant extracts used as diabetes adjuvants in Puerto Rico.

We have provided a detailed list of revisions made to improve our manuscript. We look forward to your decision. Thank you.

Michelle M. Martínez-Montemayor, Ph.D.
Associate Professor – Department of Biochemistry
Universidad Central del Caribe – School of Medicine
P.O. Box 60327, Bayamón, PR 00960
(787) 798-3001 Ext 2152